Author's response to reviews

Title: Lower Serum Triglyceride Level is a Risk Factor for In-hospital and Late Major Adverse Events in Patients with ST-Segment Elevation Myocardial Infarction Treated with Primary Percutaneous Coronary Intervention- A Cohort Study

Authors:

Yu-Tsung Cheng (smallzebra99@gmail.com)
Tsun-Jui Liu (trlu@vghtc.gov.tw)
Hui-Chin Lai (854k@vghtc.gov.tw)
Wen-Lieng Lee (wilee@vghtc.gov.tw)
Chieh-Shou Su (visitingscholarofheart@gmail.com)
Chia-Ning Liu (cutenina0420@gmail.com)
Kuo-Yang Wang (chiefkywang@gmail.com)
Hung-Yun Ho (hhy@vghtc.gov.tw)

Version: 5
Date: 2 October 2014

Author's response to reviews: see over
Dear Mr. Gilbert Tacbobo and Dear Prof. Helmut Sinzinger,

Thank you for informing me to further modify our manuscript entitled “Lower Serum Triglyceride Level is a Risk Factor for In-hospital and Late Major Adverse Events in Patients with ST-Segment Elevation Myocardial Infarction Treated with Primary Percutaneous Coronary Intervention- A Cohort Study”. According to your instructions we have modified our article to comply with your requirements. Attached please find the revised manuscript as well as the file containing “Response to Editorial Request. Please let me know if there is anything else we need to do about our article.

Best,

Kuo-Yang Wang

Chief, Department of Cardiology, Cardiovascular Center, Taichung Veterans General Hospital, Taiwan
Response to Editorial Request

1. Acknowledgements:

   - By way of a section “Acknowledgements”, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

   The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.'

   Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

   Answer: A section of “Acknowledgements” has been added to the revised manuscript (page 17, lines 1-3). Thanks again for your comments!!